ACE Weekly Report:  05-17-00

All ACE spacecraft subsystems are performing nominally. An Attitude maneuver was successfully executed on 05-15-00.  The next attitude maneuvers are scheduled for 05-22-00 and 05-26-00.

On 05-12-00 Station 66 lost its TGC, terminating the flow of telemetry to the MOC.  Station swapped TGCs and telemetry back on-line.  The remainder of the support completed without any further problems.  FOT redumped SSR slices 0-35 to recover lost data.  All SSR data successfully captured  (DR# L04289).

On 05-15-00 commanding delayed ~10 minutes due to problems with Station 66’s exciter.  Exciter control console reset by station;  successful receiver lock attained.  The remainder of the support completed without any further problems (DR# L04298).

On 05-16-00 commanding delayed ~7 minutes due to misconfiguration of Station 66’s exciter.  Exciter reset and command link successfully established (DR# G05992).

The SWICS Instrument team is adjusting the voltage of their Post-Accelerator Power Supply (PAPS).  It will take several sessions to complete the process (OCR SWX-0020).  Another session was performed on 05-16-00.
